Author's response to reviews

Title: Dynamics of growth and overweight transitions in a pediatric cohort from South India.

Authors:

Manu Raj (drmanuraj@gmail.com)
Karimassery Ramaiyer Sundaram (krsundaram@aims.amrita.edu)
Mary Paul (marypaulo@rediffmail.com)
Abish Sudhakar (abishsudhakar08@gmail.com)
Raman Krishna Kumar (rkrishnakumar@aims.amrita.edu)

Version: 2 Date: 23 July 2009

Author's response to reviews: see over
From
Dr. Manu Raj
Clinical Assistant Professor (Pediatrics)
Division of Pediatric Cardiology
Amrita Institute of Medical Sciences and Research Centre
Kochi, Kerala, India.

To
Dr. Nehme Gabriel,
Editor-in-Chief,
Nutrition Journal.

Dear Dr. Gabriel,

We have submitted a manuscript of the original work done titled ‘Dynamics of growth and overweight transitions in a pediatric cohort from South India’ to Nutrition Journal. We did not include details of the ethical clearance and consent procedure in the initially submitted version. We are revising the manuscript by adding these two sections. We are sorry for the omission from our part.

Regards,

Dr. Manu Raj